Author's response to reviews

Title: Thirty years after Alma-Ata: a systematic review of the impact of community health workers delivering curative interventions against malaria, pneumonia and diarrhoea on child mortality and morbidity in sub-Saharan Africa

Authors:

Jason B Christopher (drjchristopher@gmail.com)
Alex Le May (alex.lemay@gmail.com)
Simon Lewin (Simon.Lewin@nokc.no)
David A Ross (david.ross@lshtm.ac.uk)

Version: 5 Date: 4 October 2011

Author's response to reviews:

The author list has been made the same in the manuscript and submission system- removing professional titles.
Affiliations have been give using superscripts as requested. Corresponding author has been designated.
The 3 figures have been removed and entered into the text of the manuscript as advised. The text within the tables are single-spaced, not double spaced as in the rest of the submission.
The sentence - All authors have read and approved the final manuscript has been added to Author Contributions.
I have checked all the references, now using ()s for dates of website access.
I have checked the manuscript for typographical areas.